

April 29, 2024

Dear editor,

It is my pleasure to reply to you about the feedback received by the reviewers of our manuscript number 91214.

We received comments from one reviewer. Those were:

1. The title should be changed, the same title as the original article is inappropriate.
2. In the third teaching capsule, the author think patient under dialysis treatment face deteriorating health conditions, it's a little arbitrary. It's better for author to provide evidence or related article.
3. The fourth and fifth teaching capsule express the similar point of view, could be combined.
4. The seventh teaching capsule is a little verbose, it's better to simplify.

Regarding the first comment, the reviewer is right in saying this editorial should have a different name. We included it in the revised manuscript, and specified it is an editorial of a specific article below the title.

Regarding the second comment, it is widely known that patients in dialysis deteriorate faster than a healthy individual, or even faster than transplanted patients. Nonetheless, the phrase was not referring to the deteriorating conditions caused by dialysis, but the deteriorating conditions caused by other health conditions or social determinants of health. We improved the writing to improve understanding, and included a reference for that.

Regarding the third comment, we didn't want to express similar teachings in fourth and fifth teaching capsules. We improved the writing to improve understanding.

Finally, we edited the writing of the seventh teaching capsule.

Thank you for this opportunity, sincerely yours,

Fernando Gonzalez